

HHS Public Access

Life Sci Space Res (Amst). Author manuscript; available in PMC 2018 January 17.

Published in final edited form as:

Author manuscript

Life Sci Space Res (Amst). 2016 February ; 8: 38–51. doi:10.1016/j.lssr.2016.02.001.

Galactic cosmic ray simulation at the NASA Space Radiation Laboratory

John W. Norbury^{a,*}, Walter Schimmerling^{b,c}, Tony C. Slaba^a, Edouard I. Azzam^d, Francis F. Badavi^e, Giorgio Baiocco^f, Eric Benton^g, Veronica Bindi^h, Eleanor A. Blakelyⁱ, Steve R. Blattnig^a, David A. Boothman^j, Thomas B. Borak^k, Richard A. Britten^l, Stan Curtis^m, Michael Dingfelder^b, Marco Duranteⁿ, William S. Dynan^o, Amelia J. Eisch^j, S. Robin Elgart^p, Dudley T. Goodhead^q, Peter M. Guida^r, Lawrence H. Heilbronn^s, Christine E. Hellweg^t, Janice L. Huff^c, Amy Kronenbergⁱ, Chiara La Tessa^r, Derek I. Lowenstein^r, Jack Millerⁱ, Takashi Morita^u, Livio Narici^v, Gregory A. Nelson^w, Ryan B. Norman^a, Andrea Ottolenghi^f, Zarana S. Patel^p, Guenther Reitz^t, Adam Rusek^r, Ann-Sofie Schreurs^x, Lisa A. Scott-Carnell^a, Edward Semones^p, Jerry W. Shay^j, Vyacheslav A. Shurshakov^y, Lembit Sihver^{z,aa}, Lisa C. Simonsen^a, Michael D. Story^j, Mitchell S. Turker^{ab}, Yukio Uchihori^{ac}, Jacqueline Williams^{ad}, and Cary J. Zeitlin^{ae}

^aNASA Langley Research Center, Hampton, VA 23681, USA

^bEast Carolina University, Greenville, NC 27858, USA

^cUniversities Space Research Association, Houston, TX 77058, USA

^dRutgers University, Newark, NJ 07101, USA

^eOld Dominion University, Norfolk, VA 23529, USA

^fDepartment of Physics, University of Pavia, 27100, Pavia, Italy

^gOklahoma State University, Stillwater, OK 74074, USA

^hUniversity of Hawaii – Manoa, Honolulu, HI, 96822, USA

ⁱLawrence Berkeley National Laboratory, Berkeley, CA 94720, USA

^jUniversity of Texas Southwestern Medical Center, Dallas, TX 75390, USA

^kColorado State University, Fort Collins, CO 80523, USA

^IEastern Virginia Medical School, Norfolk, VA 23507, USA

^m11771 Sunset Ave. NE, Bainbridge Island, WA 98110, USA

ⁿGSI Helmholtz Center for Heavy Ion Research, 64291 Darmstadt, Germany

°Emory University, Atlanta, GA 30322, USA

PNASA Johnson Space Center, Houston, TX 77058, USA

^qMedical Research Council, Harwell, Didcot OX11 0RD, United Kingdom

^rBrookhaven National Laboratory, Upton, NY 11973, USA

^{*}Corresponding author. Tel.: +1 757 864 1480; fax: +1 757 864 8911. john.w.norbury@nasa.gov (J.W. Norbury).

^sUniversity of Tennessee, Knoxville, TN 37996, USA
^tGerman Aerospace Center, 51147 Cologne, Germany
^uOsaka City University, Osaka 545-8585, Japan
^vUniversity of Rome Tor Vergata & INFN, 00133 Rome, Italy
^wLoma Linda University, Loma Linda, CA 92354, USA
^xNASA Ames Research Center, Moffett Field, CA 94035, USA
^yInstitute of Biomedical Problems, Moscow 123007, Russia
^zTechnische Universität Wien – Atominstitut, 1020 Vienna, Austria
^{aa}EBG MedAustron GmbH, 2700 Wiener Neustadt, Austria
^{ab}Oregon Health and Science University, Portland, OR 97239, USA
^{ac}National Institute of Radiological Sciences, Chiba 263-8555, Japan
^{ad}University of Rochester Medical Center, Rochester, NY 14642, USA

Abstract

Most accelerator-based space radiation experiments have been performed with single ion beams at fixed energies. However, the space radiation environment consists of a wide variety of ion species with a continuous range of energies. Due to recent developments in beam switching technology implemented at the NASA Space Radiation Laboratory (NSRL) at Brookhaven National Laboratory (BNL), it is now possible to rapidly switch ion species and energies, allowing for the possibility to more realistically simulate the actual radiation environment found in space. The present paper discusses a variety of issues related to implementation of galactic cosmic ray (GCR) simulation at NSRL, especially for experiments in radiobiology. Advantages and disadvantages of different approaches to developing a GCR simulator are presented. In addition, issues common to both GCR simulation and single beam experiments are compared to issues unique to GCR simulation studies. A set of conclusions is presented as well as a discussion of the technical implementation of GCR simulation.

Keywords

Space radiation; Galactic cosmic ray simulation

1. Introduction

The health effects of space radiation on astronauts represent a major limiting factor for longduration human space missions beyond Low Earth Orbit (LEO) (Durante, 2014). Beyond LEO, the most important sources of space radiation consist of galactic cosmic rays and Solar Particle Events (SPE). GCR nuclei of average energy can penetrate a substantial thickness of materials, on the order of 10s to 100s of centimeters of water or aluminum. If a nuclear

interaction between a primary GCR ion and a target nucleus occurs, the lighter secondary products will lose energy at a lower rate, and therefore will be able to penetrate even further. For this reason, it is not possible to provide sufficient shielding material to fully absorb all types of radiation in space. In addition, the relative biological effectiveness of nuclei will change as a function of depth of penetration, because the composition and energy of the nuclei changes due to atomic and nuclear interactions. The Linear Energy Transfer (LET) of each nucleus also changes as it loses energy and slows down inside the material being penetrated.

The major GCR particle types include hydrogen (H), helium (He), carbon (C), oxygen (O), neon (Ne), silicon (Si), calcium (Ca), and iron (Fe). The energy spectra of all GCR particles are very broad with the region extending from approximately 10 MeV/n to 50 GeV/n being of primary importance to space applications (Grahn, 1973; Slaba and Blattnig, 2014; Durante and Cucinotta, 2011). GCR exposures occur at low fluence rate, with each cell in an astronaut's body being "traversed by a proton about every three days, helium nuclei once every few weeks, and high atomic number (Z) and energy (HZE) nuclei about once every few months" (NASA, 2015a). The cells are not traversed at random and the traversals are not statistically independent. The traversal of a cell nucleus usually correlates with the simultaneous traversal of very large numbers of additional cell nuclei (on the order of 10^9) in the tissue along the track of the same particle. These fluence rates correspond to tissue doses or effective dose-rates of about 0.3–0.6 mGy/day and 1–1.8 mSv/day, respectively. However, the use of absorbed dose (or dose-rate) is misleading, because the energy lost by each incident particle is deposited in a highly non-uniform way, both physically and temporally. Dose-rate effects may best be understood in terms of particle fluence rate (commonly referred to as particle flux); any such effects will be dependent on the end-point considered, and the time constants of the chemical kinetics involved.

SPEs consist primarily of protons and, much like GCR, have a broad energy spectrum with the energy region of most importance to human space flight extending out to a few hundred MeV. The SPE spectra include much smaller components of helium and heavy nuclei. The shapes of the energy spectra, as well as the total fluence, vary considerably from event to event. Over the course of an SPE, dose-rates can fluctuate between 0–100 mGy/hr inside the protection of a vehicle. SPE dose-rates can also differ by several-fold between tissue sites because of the variable energy spectra of the protons or other nuclei. Similar to the case for GCR, the use of dose or dose-rate to characterize protons in space can be misleading from a biological point of view, except in cases where the proton fluence is high enough to ensure that the target organism has been irradiated uniformly. Note, however, that statements concerning dose and dose-rate for SPE are also dependent on the space vehicle or space habitat analyzed and the SPE spectrum chosen. There are large variations across both of those variables.

Energy deposition in biomolecules, cells, and tissues is distinct when comparing protons and HZE nuclei to common forms of terrestrial radiation. For the particles comprising space radiation, energy deposition is highly localized along the trajectory of each particle with lateral transport of energetic electrons (delta-rays) away from the nuclei's path. The rate of energy deposition per unit length of a particle trajectory is described as LET. The unit

generally used in radiobiology for LET is the kilo-electron volt per micrometer, or keV/µm. The LET of charged particles changes as a function of particle velocity, β , or kinetic energy, and its charge, *Z*, approximately in proportion to Z^2/β^2 . As the velocity (or energy) of a particle increases, the LET decreases to a minimum near a velocity of approximately 90% of the speed of light; at higher energies the LET increases very slowly due to relativistic effects. High-energy charged particles lose energy when they traverse any material. As they slow down, the LET increases to a maximum and then very rapidly decreases to zero. The low-energy maximum in the LET occurs very close to the point where the charged particle loses its remaining energy and stops. Nuclear fragmentation and other nuclear interactions, including projectile fragmentation of the primary ion and target fragmentation, target fragmentation, including secondary neutron production, introduces an additional high LET component into the radiation field.

Space radiation risks of concern to NASA are carcinogenesis (increased risk to fatal cancers), acute (in-flight) and late (i.e. after a mission) risks to the central nervous system (CNS), degenerative tissue risks such as cardiovascular disease, and acute radiation syndromes. For cancer and acute risk estimates, human epidemiology data with gamma-ray and X-ray exposures play a key role in risk estimation models. Acute risks are a concern for SPE, while cancer, CNS, and cardiovascular risks, etc., are a concern for both GCR and SPE. The current model of cancer risks used by NASA, NSCR 2012 (Cucinotta et al., 2013) scales cancer incidence or mortality rates estimated from epidemiology data to the effects for the low dose-rates and radiation types in space using a dose- and dose-rate effectiveness factor (DDREF) and radiation quality factor, respectively. There are large uncertainties in this model, which, in order of decreasing importance are as follows: the radiation quality factors, dose and dose-rate dependencies, the transfer of risk across populations, the determination of space radiation organ exposures, and the various errors in human data sources. In addition, there are uncertainties related to the underlying assumptions of the model due to possible qualitative differences between high- and low-LET radiations, the validity of the assumptions of linearity and additivity of effects for different radiation components, and the possible synergistic risks from other flight factors on radiation risks. Because solar protons are largely low LET, and the proton fluences during SPE's can be highly variable, the predominant uncertainty for acute risk estimates is related to the understanding of dose-rate effects.

Of greatest concern are the effects of GCR from hydrogen to nickel in the approximate energy range from 10 MeV/n to 50 GeV/n (Slaba and Blattnig, 2014; Durante and Cucinotta, 2011). Space-based studies to investigate the physics, biology, and clinically evident health consequences of space radiation suffer from severe limitations. Among them are the necessary choice of minimal radiation exposure to spacecraft crews; the small number of humans with space experience; restrictions on mass, volume and power available for equipment and maintenance of laboratory animals; and the lack of real-time access to experiments by investigators. For these and similar reasons, NASA recognized early in the development of space radiation research that statistically significant data with the accuracy required for proper risk management could only be obtained by the use of high-energy, ion beams at ground-based accelerators. Consequently, there have been research programs

concentrating on understanding the effects of heavy ions conducted at several particle accelerators around the world (Sihver, 2008). The first acceleration of relativistic heavy ion beams occurred at Princeton in 1971 (White et al., 1971; Isaila et al., 1972), followed by the BEVALAC at Lawrence Berkeley National Laboratory (LBNL), the Helmholtzzentrum für Schwerionenforschung (Helmholtz Center for Heavy Ion Research, formerly Gesellschaft für Schwerionenforschung, GSI), the NASA Space Radiation Laboratory (NSRL) at Brookhaven National Laboratory (BNL), the Heavy Ion Medical Accelerator (HIMAC) at the National Institute of Radiological Sciences (NIRS) in Chiba, Japan, and additional accelerator facilities in Europe and Asia. An active program is currently underway at NSRL, and a future program at higher energy, above 10 GeV/n, is planned at the Facility for Antiproton and Ion Research (FAIR) at the Helmholtzzentrum für Schwerionenforschung (Durante et al., 2007; Stohlker et al., 2015; Sihver, 2008; Chattopadhyay, 2014; Heuser, 2013).

To date, typical ground-based experiments have involved the use of single ion beams, with or without degraders, at fixed energies. However, accelerator technology has developed to the stage where it has now become feasible to accelerate a variety of beams with differing energies, with beam switching times sufficiently small (less than two minutes) that allow approximate reproduction of parts of the GCR spectrum in ground-based facilities (Slaba et al., 2015a, 2016) compressed into a high dose-rate exposure, or beams consisting of various ions simultaneously. This will allow researchers to simulate a mixed field in tissue to investigate the combined biological effect of different LET irradiations. This new experimental capability better simulates the space environment and allows for ground based experiments to replicate the important biological effects of a mixed radiation spectrum, e.g. the field of light ions (defined as isotopes of hydrogen and helium), that are major contributors to space radiation dose equivalent along with representative heavy ions (Norbury and Slaba, 2014; Walker et al., 2013).

It is useful to identify the main reasons for considering GCR simulation. In preparation for sending astronauts on long duration missions into space, it is necessary to simulate the same overall environment in ground-based facilities. Of course, the practical implementation of GCR simulation will be different from the real space environment in terms of the particle spectrum, dose and dose-rate. While the implementation of GCR simulation will only be an approximation to the true situation in space, it is a great improvement over the single ion exposure now widely in use. Therefore, current approaches would be complemented with single beams to develop radiobiological mechanistic understanding and phenomenological relationships. Knowing what harmful health effects occur leads to the development of appropriate countermeasures. This reasoning is a biological end point phenomenological approach and does not rely entirely on standard radiobiological issues such as cancer mechanisms, predictive models or risk analysis. However, if extrapolated to its full extent, this approach will impose a set of experimental conditions that will be needed in order to eliminate or isolate the risks from GCR alone as well as in combination with other space "toxicities", such as microgravity or a restrictive diet. Nonetheless, determining risk from the space environment is necessary if GCR exposure alone leads to such risks as increased carcinogenesis, cognitive dysfunction, and cardiovascular problems, since efforts can then be targeted towards risk reduction. Indeed, the recent NASA Research Announcement (NRA)

(NASA, 2015a) gives an indication of what are expected of the results of GCR simulation: "Ultimately mixed field studies must inform risk model predictions, biomarker discovery and validation, and countermeasure identification and validation across all risk areas" (NASA, 2015a; Appendix D, p. 16). Thus, results from mixed field studies must ultimately find their way into risk models. The mechanisms and repair or recovery pathways seen after exposure to single ions may be quite different, particularly with dose-rate. It is important to know how relevant space radiation doses and dose-rates of mixed field exposure damage cells, and GCR simulation will make that feasible. It can be challenging, however, to design experiments with adequate statistical power for very low dose, low dose-rate studies.

Finally, GCR simulation can benefit from studies from the particle physics community, such as better measurements of nuclear cross sections at different energies, a precise knowledge of the GCR flux for different species at all energies throughout the solar cycle, and developing more effective solutions to reproduce a particle dose more similar to the one of GCR. For example, the Alpha Magnetic Spectrometer AMS-02 (Aguilar et al., 2015) on the International Space Station (ISS) is providing precision flux measurements of all cosmic rays species during the ascending and descending phase of the solar cycle that can further constrain the GCR simulation model and improve the final results.

The outline of the rest of this paper is as follows: Section 2 presents a discussion of previous studies of GCR simulation after introducing the two concepts of external field versus local tissue field approaches to GCR simulation. Section 3 discusses GCR simulation from the NASA perspective. This is followed by Section 4 which highlights the importance of continuing single beam experiments to complement GCR simulation experiments. Sections 5 and 6 are devoted to some on-going important issues that arise when implementing a GCR simulator at an accelerator laboratory, such as NSRL: Section 5 specifically addresses issues common to both single beam experiments and GCR simulation experiments, and Section 6 addresses issues that are unique to GCR simulation alone. Section 7 branches out to other applications of GCR simulator at NSRL. Finally, Section 9 presents a summary and some conclusions.

2. Previous studies

Before proceeding to the full discussion of GCR simulation, it is worthwhile to summarize previous studies of this subject. This section starts with a background discussion of the external field versus local tissue field methods, prior to delving into the previous studies, which include several workshops and two journal papers.

2.1. External field approach versus local tissue field approach to GCR simulation

A topic that sometimes causes confusion comes when specifying the actual ions and energies that should be provided by an accelerator laboratory GCR simulator. The answer is not as simple as providing the same ions and energies seen in space, because it would be almost impossible for an accelerator to provide a continuous energy spectrum for each ion. There are two approaches to GCR simulation called the "external field" approach and the "local tissue field" approach that have been studied thus far (Slaba et al., 2015a, 2016; Kim

et al., 2015). The external field approach to GCR simulation aims to directly simulate the GCR spectrum which is external to a spacecraft shield. This external spectrum is significantly modified upon passage through the spacecraft, with many more neutrons and light ions of reduced energy present in the interior of the spacecraft. It is this modified interior spectrum to which an astronaut is actually exposed. Simulating the external GCR spectrum is called the external field approach, while simulating the spectrum occurring within an astronaut is called the local tissue field approach. In implementing GCR simulation at a ground-based accelerator, the required particle energies are much lower for the local tissue field approach. Because of the relatively low energy of NSRL, the local tissue field approach may provide a more flexible and feasible solution at this time. Both methods have their own set of limitations. The external field approach might be more suitable at higher energy machines, such as FAIR (Durante et al., 2007). All of these issues are discussed extensively in the accompanying article by Slaba et al. (2016) and the paper by Kim et al. (2015).

2.2. Workshops

Several workshops have recently focused on GCR simulation at NSRL. The first workshop was held at NASA Langley Research Center in Hampton, Virginia in October, 2014. Participants included fifteen physicists and radiobiologists from a variety of institutions including NSRL, and all participants are coauthors of this paper. The second workshop was held in January, 2015 as part of the NASA Space Radiation Investigators' Workshop in Galveston, Texas. This second workshop included about fifty participants from a much broader variety of institutions, and again many of these participants are also coauthors of this paper. A third workshop was held at the Radiation Research Society meeting in Weston, Florida in September, 2015. In order to promote a broad international discussion, a collaboration was developed, including the many coauthors of this paper. This broad input from the international community was also the subject of several presentations at international conferences (Slaba et al., 2015b; Norbury et al., 2015a, 2015b, 2015c).

A list of issues and questions about the necessity and implementation of GCR simulation were raised at the various workshops mentioned above. For example, is GCR simulation really needed or wanted by space radiobiologists? If so, what scientific questions would GCR simulation help address? What knowledge gaps would GCR simulation help close? And how in particular would GCR simulation be useful? In answer to these questions, space radiobiologists were generally very enthusiastic about the availability of ground-based GCR simulation, although there were numerous concerns about the practicality of its implementation in regards to animal housing, personnel, financial costs, and duplicative efforts across laboratories. To this end, a call was made for more unified standards in radiation parameters, and for more aggressive tissue/organ sharing from current and future NSRL experiments, an effort that is already underway (NASA, 2015b). Dosimetry questions were also raised. How would dosimetry on a GCR simulation combination of beams be performed, and how would one measure the exposure dose of individual beams in the mixed field? In answer to these questions, the NSRL staff stated that scintillators would be used for low dose experiments and there would be no serious dosimetry issues in changing from high to low dose. Questions about the external vs. local tissue field issues were also raised,

Page 8

including: Would a combination of both the local tissue field approach and the external field approach be possible, with perhaps a shield or water phantom positioned downstream? It was discussed that from a biological perspective, this dual field approach may be better. Numerous other ideas and issues were also generated. For example, it was pointed out that the idea of GCR simulation is not new and the idea has been around for about twenty years. The hope of microgravity simulations was raised, as was the potential for delivery of sequential beams (i.e. delivery of one beam after another) versus a full mixed beam field, with a mixture of beams delivered at the same time (perhaps all at the same rigidity).

2.3. Study of Slaba et al

A detailed implementation study of GCR simulation has recently been completed by Slaba et al. (2015a, 2016), which focused on specifying the reference field and spelling out strategies for selecting the beams at NSRL. A detailed GCR simulation reference field in terms of ion species, beam energies and dose can be found in Tables A1, A2, A3 of that paper. The study appears as a companion paper in the present journal volume and is divided into three main areas.

First, the issue was considered of whether to simulate the free space GCR environment (external to a spacecraft) versus simulating the GCR field induced at local tissue sites, given current accelerator facility constraints. The latter option – simulation of local tissue GCR field – was chosen as being the most feasible implementation for GCR simulation at NSRL due to the maximum energy available, which at the upgraded NSRL will be 1. 5 GeV/ nucleon for heavy ions and 4.0 GeV for protons, versus the currently available energies of 1. 0 GeV/nucleon and 2.5 GeV, respectively. (Note that issues of the beam traversing mouse and cell containers etc. do not significantly alter these two basic approaches to GCR simulation.)

Second, the question was addressed concerning the need for multiple GCR simulation reference fields versus a single reference field. Obviously, a single field would be preferred from a cost and convenience perspective. It was found that variations in shielding configurations, variable GCR fluence due to solar activity, and tissue location within the body are smaller than or approximately equal to uncertainties associated with modeling the free space GCR environment, transport models and experimental designs. This allowed for the selection of a single GCR simulation field which was chosen to represent the spectrum induced in female blood forming organs (BFO) behind 20 g/cm² of aluminum shielding during solar minimum. (Note that it is this GCR simulation reference field that will be seen directly by astronauts in a typical space vehicle and therefore it is this field that is being proposed for GCR simulation lab experiments at NSRL.)

Third, a preliminary beam selection strategy was considered. As emphasized in previous work (Norbury and Slaba, 2014; Walker et al., 2013), isotopes of hydrogen and helium, collectively called light ions, make a significant (sometimes dominant) contribution to dose equivalent for realistic thick shielding used in current and future vehicle designs. Therefore, hydrogen and helium needed to be considered separately from heavy ions in the GCR simulation beam selection strategy and the hydrogen and helium energy spectra are represented directly. For heavy ions, a selection strategy based on reproducing the LET

spectra was used. This method also was shown to very well predict track structure spectra, based on the track structure parameter, Z^{*2}/β^2 .

In addition to those three main sections, Slaba et al. (2015a, 2016) briefly discussed some of the disadvantages of the proposed method, such as the lack of inclusion of neutrons, pions, and associated electromagnetic (EM) cascades. Neutrons are not present in the external GCR spectrum, but are produced copiously when GCR particles undergo nuclear reactions with spacecraft shielding material. These secondary neutrons impinge on astronauts' bodies. If one were using the external field approach to GCR simulation, then external beams would hit aluminum targets with secondary neutrons produced on the other side of the target. The local tissue field approach, which simulates the beams directly hitting tissue, is unable to include the neutron-induced charged particle field because neutron beams are unavailable in conjunction with ion beams of sufficient energy. Pions and electromagnetic cascades (electrons, muons, photons) are also secondary particles that are not present in the local tissue field approach.

Slaba et al. (2015a, 2016) further considered avoidance of Bragg peaks in animal experiments, providing a clear example of how biological constraints can be included in the simulator design process. Specifically, they propose that lower energy (stopping particles) should be represented but hot spots (Bragg peaks) should be avoided. If Bragg peaks are present in different places in animals in different experiments, experimental reproducibility will be difficult, if not impossible. This was later addressed in a revised strategy for hydrogen and helium ions (Slaba et al., 2015a, 2016), which specifically avoided Bragg peaks in experiments with mice. A modified approach would be needed for any larger species. This strategy could be extended to heavy ions, but would greatly increase the burden on the facility for fast ion and energy switching. The idea would be to test all (or maybe partial) implementation of the GCR simulator at NSRL and also answer some of the major radiobiology and logistical questions related to the simulator (assuming they can be clearly identified). It is reasonable to note that avoidance of Bragg peaks in the experimental designs (leading to better reproducibility) produces a difference to the actual situation in space, where GCR particles are always stopping in tissue and target materials. However, because the space environment involves a large variety of nuclei at a large variety of energies, these hot spots gradually coalesce into a spread out Bragg peak, exactly as implemented in the local tissue field approach (Slaba et al., 2015a, 2016). In the local tissue field approach, each beam will later transform into a variety of particles at different energies after interacting with tissue material.

2.4. Study of Kim et al

Recently, Kim et al. (2015) published the results of an extensive simulation using the stochastic transport code, GERM (GCR Event Risk Model), to define a GCR reference field using 9 HZE particle beam–energy combinations, each with a unique absorber thickness to provide fragmentation and 10 or more energies of proton and He beams. Their kinetics model of HZE particle hit probabilities suggests that experimental exposures of several weeks will be needed to avoid high fluence rate artifacts, which places limitations on the experiments that may be feasible.

Nuclear reactions in absorbers or tissue equivalent materials, as well as in tissues of rodents and holders, may also emit neutrons. Kim et al. (2015) make four important points: (1) neutrons may be more important on Mars than in transit; (2) high energy neutrons will not be very effective; (3) the number of neutrons produced in heavy ion fragmentation is about the same as the number of protons, so it should be relatively straightforward to multiply the proton flux by a factor to account for neutron-induced nuclear reactions; (4) neutrons are produced by incident GCR particles – there is not a significant number of free neutrons in space, so there is no need to generate extra neutrons in a GCR simulator which uses the external field approach.

Kim et al. (2015) also provided a detailed discussion of the importance of biological relaxation times in planning fluence-rate and length of exposure. These included the turnover times of different tissue types and the distinct mechanisms for targeted and non-targeted effects in cancer risk. In addition, slowly and rapidly dividing tissues, as well as abscopal effects, could require different optimal chronic exposure times. The authors also conclude that, for CNS risks, changes in cognition and behavior, may be through multiple mechanisms that can culminate in acute and late outcomes, including synaptic changes. They specifically pointed out that "synapse formation, stabilization, and decay have a fast actin-dependent component (less than one day) and a slow plasticity-dependent component (days to years)". Importantly, the average lifetime of synapses varies not only in different regions of the brain, but also in comparison between mice or rats and humans.

The authors go on to consider practical aspects of setting up experiments that require longterm or split irradiations to take advantage of a GCR simulation facility. They conclude that it is essential to base its use on a solid scientific basis, and suggest several areas of interest, including the possibility that synergistic interactions of particles of different radiation qualities may lead to differences in biological responses for a mixed-field of particles of varying track structure.

3. NASA requirements

This section examines GCR simulation from the NASA perspective, including GCR simulation study requests in recent research calls. In March, 2015, NASA issued an NRA (NASA, 2015a) that set forth the requirements for development of GCR simulation. These requirements are quoted here:

"Study formulations should consider that optimized combinations of particle types and energies may differ across risk areas (e.g., cancer vs. CNS) and across model validation testing.

- How many different particles (including particle types and energies) are sufficient to adequately simulate the biological and health consequences of GCR across risk areas? What is the time scale (dose-rate and duration) and order of delivery of particle types?
- What mixed radiation field, including protons, helium and heavier nuclei, provides the most stringent tests of model predictions?

• Is there an intermediate capability that may provide a similar statistical power for determining the validation of radiation models?

In addressing the above questions, the design and implementation of critical experiments are needed to support the definition of end user capabilities. Topics may include, but are not limited to, the following:

- Studies that implement LET-sensitive experiments that test hypotheses, based on existing experimental data with single ion exposures, about how experimental endpoints may change in response to a mixed field chronic exposure.
- Studies that implement geometry/time-sensitive experiments to scale cell and animal model results to humans at very low doses, supporting the extrapolation of simulator experimental exposure schemes to space relevant particle fluxes.
- Studies to determine whether the effects of different particles, or particles delivered at different times, depend on the order in which they are delivered. Studies should consider the effects of the order of heavier ions (high LET) on cells/animals previously exposed to protons (low LET).
- Studies that implement time-scale-sensitive experiments that consider the different lifetime scale of animals and humans. At low doses, these effects may not be easily separated from the effects of geometry (e.g., number of cells, or fraction of tissue or organ responding to the radiation)."

The need to address issues related to effective simulation of the space radiation environment and knowledge gaps identified in order to provide accurate risk assessment are outlined in the NASA Human Research Program's strategic research plan (NASA, 2015c). This document covers the four major risk areas of radiation carcinogenesis, acute and late CNS effects, cardiovascular/degenerative tissue effects and acute radiation syndromes due to SPEs. For example, the degenerative tissue (non-cancer or non-CNS) effects such as cardiovascular disease, cataracts, and others including digestive and respiratory diseases, are documented in epidemiology as well as in experimental studies following exposures to terrestrial sources of ionizing radiation (e.g., gamma rays and X-rays) (Little et al., 2012). However, low dose thresholds, dose-rate and radiation quality effects, as well as mechanisms and major risk pathways, are not well-characterized. Likewise, for cancer risk assessment, the effect of protracted exposures simulating the dose-rates found in the space environment on cancer incidence and tumor spectrum is a major open question. GCR simulation studies at NSRL will contribute to the body of evidence required to close these knowledge gaps and provide critical data for accurate modeling of these risks.

4. Single beam experiments

While there is interest in pursuing GCR simulation studies, it should be understood that such a program would never completely replace single beam experiments. Also, single beam experiments will be needed as a reference to the multiple beams used in GCR simulation. A healthy balance of both GCR simulation and single beam studies should be pursued. The original idea behind doing experiments at NSRL was to elucidate cancer and other disease mechanisms. With the multiple beam studies in GCR simulation, it becomes unclear as to

which beam causes a specific outcome. Comparison to single beam experiments is therefore considered to still be very important, and may remain necessary for the understanding of the basic interaction mechanisms between radiation and biological models, from cells to tissues to whole organisms. It is evident that single beam studies are needed as a reference to observe additive or even synergistic effects of simultaneously acting different LET irradiations. They are also useful in deciphering the extent of any mitigating effects that a chronic low LET radiation exposure may exert against damaging biochemical changes induced by a subsequent exposure to high LET radiation. Conversely, it is possible that in some instances a chronic low LET exposure may exert potentiating effects that will alter the effect of a subsequent traversal by a heavy ion. At high doses in charged particle therapy, non-linear additive models are used to predict tumor cell killing in mixed radiation fields generated by projectile fragmentation (Loeffler and Durante, 2013). Exposures of biological systems to energetic heavy ions behind thick shields, such as might realistically exist in spacecraft or in a habitat on a planetary surface, also generally produce data compatible with additive functions, and the beam quality can be represented by the dose-averaged LET (Herr et al., 2015) or associated microdosimetric quantities (ICRP, 2013).

The GCR simulator fills an important gap in NASA's approach to understanding space radiation risk, since NASA wants to establish risk in general, not just for individual ions or protons. In addition, countermeasure testing and biomarker validation in a mixed field will be an important capability in the near future. It is important that validation of countermeasures and biomarkers be performed in a mixed field more closely representing the true space environment. Studies have shown that countermeasures that are efficacious in a gamma or neutron field do not exhibit the same efficacy under similar conditions in a mixed gamma/neutron field suggesting the mechanisms may be quite different for single field exposure compared to a mixed field, particularly when radiation quality is considered (Cary et al., 2012). It is also important to remember that experiments at NSRL are intended to place risk assessment on a solid basis of fundamental science, and are therefore focused on establishing the biological mechanisms underlying the health effects of space radiation. Single beam studies, in addition to GCR simulation, will therefore always be needed. In addition, the selected particle types and energies chosen in the mixed field studies (e.g., 1 GeV protons, 600 MeV/n Fe) should be relatable back to the vast amount of existing radiobiological experimental data collected over the past several decades. Likewise, future single beam experiments should consider using particle types, energies, and exposures consistent with GCR simulator species to enhance interpretation of experimental results.

5. Issues common to both GCR simulation and single beam experiments

Some of the issues raised at the workshops mentioned previously and subsequently discussed by the broader community will now be considered in more detail. Many issues were raised about space radiation experiments in general, as opposed to problems particular to GCR simulation. It is therefore worthwhile to break the further discussion into two main areas. The first discusses issues that apply to all space radiation experiments, including both GCR simulation experiments as well as the current paradigm of single beam experiments discussed in the previous section, and the second discusses issues that apply to implementation of GCR simulation alone.

5.1. Low dose-rate

According to the recent NRA (NASA, 2015a), each cell in an astronaut's body in deep space is traversed by a proton every three days, by a He nucleus every few weeks, and by a heavy ion every few months. NASA states that "these fluence rates correspond to tissue doses or effective dose-rates of about 0.3-0.6 mGy/day and 1-1.8 mSv/day". Dose-rates available at NSRL are up to 15 Gy/min for the 20×20 cm² beam configuration and about 0.5 Gy/min for the 60×60 cm² configuration (NASA, 2015a). As an example of lower dose rates available, one investigator regularly uses 0.2 cGy/min for 10-50 cGy total dose. Due to facility and running time constraints, the majority of ground-based radiobiology beam experiments have used much higher dose-rates than those that occur in space. However, as is well documented, biological mechanisms are profoundly different following high versus low and very low doses, at least for sparsely ionizing X- or gamma-rays, and there is no reason to believe that this may not also be true at equivalent dose-rates for sparsely ionizing high energy protons and alpha particles (Hall and Giaccia, 2012). Therefore, low dose-rate experiments, such as extended fractionations, need to be performed. This will, of course, require greater cooperation between experimental teams. For example, the medical facility staff at Brookhaven National Lab could perform animal irradiations at fractionated intervals over a long period of time without requiring the presence of principal investigators on site. A program of extended low dose-rate experiments will require a larger commitment for beam time and NASA fiscal support. Efforts to facilitate tissue sharing across multiple principal investigators and experimental endpoints will greatly enhance the scientific return on investment.

Specifically, three critical issues – feasibility of low dose-rate, availability, and cost of beam time – are important challenges to be solved, particularly since the latter two are critical. The following possibilities should be considered to address these challenges: 1) using a controlled large angle irradiation as a piggy-back experiment; 2) creating multiple, switchable beam output in the same experimental room (in order to "multiplex" irradiation, to decrease costs and increase availability); and 3) creating multiple simple experimental rooms, each using one of the above multiplexed beams. Each of these possibilities are accompanied by additional challenges and costs, emphasizing the importance of identifying more long-term, feasible approaches to practically implementing ground-based GCR simulation. Cancer is a disease that progresses over a long period of time; therefore, one might measure the effect of radiation after the initiating event in order to look at variations in tumor progression. One has to consider progression as well as initiation. The risk for carcinogenesis when oncogenically progressed cells are irradiated is much higher when given acutely. It is unclear whether this is the case at low dose-rate and is likely very specific to the -omic alteration that renders a cell as oncogenically progressed.

Experiments at NSRL are not necessarily designed to input directly into risk models, but rather to study the basic biological pathways of carcinogenesis. These limitations are not isolated to cancer research; the bones, eyes, CNS, cardiovascular, and other systems may incur substantial and short timeframe damage and functional degradation in the compressed timeframe of current lab experiments which may or may not be accurate for space. Data from human exposures for cardiovascular effects note varying results with fractionated

exposures in humans (low-LET) and the inverse dose rate effect seen with high LET ²⁵²Cf neutron exposures for lung and mammary carcinogenesis (Ullrich et al., 1977) suggest that effects of dose-rate for high-LET radiation must be considered (Delgado et al., 2014).

5.2. Overlapping tracks

One effect of delivering beams at high dose-rate is that heavy ion tracks can overlap (Curtis, 1994) when passing through cellular material. In contrast, with the actual low dose-rates experienced in space, tracks will not overlap, except with respect to large target volumes and/or long biological relaxation times. Note that overlaps depend on the volume of interest (e.g. DNA versus the full range of delta rays even out to ~cm for 1 GeV/n) and the time of interest, such as physico-chemical times of 10^{-9} sec versus years for some long-term biological effects. Therefore, a possible concern is that GCR simulation won't help to improve NASA risk models. In order for biological experiments to avoid the overlap effect, experiments at low dose-rate would minimize multi-track hits per cell nucleus per exposure, and thus bring these experiments closer to the space radiation environment.

5.3. Dose

The concept of "dose-response" is a fundamental concept in radiobiology, where one wishes to determine a certain type of biological or other response of the system to the amount of dose delivered. Often this is represented graphically with the dose plotted on the horizontal axis and the response plotted on the vertical axis. It is important that a GCR simulator is able to represent the dose experienced in space, so that the response that one observes corresponds to the response that would occur on a particular mission such as a trip to Mars. Recent measurements (Zeitlin et al., 2013) of the space radiation environment in transit to Mars yield a dose equivalent rate of 1.84 ± 0.33 mSv/day using ICRP60 quality factors (0.48 \pm 0.08 mGy/day). For a 180-day cruise to Mars, this corresponds to a dose of 86 mGy and 172 mGy for the roundtrip journey. Adding a 500-day Mars surface exposure of about 105 mGy (Hassler et al., 2014; Kerr, 2013) yields a total dose of about 277 mGy, which is roughly 1 Sv for a 3-year Mars mission. The recent NRA (NASA, 2015a) requires experiments with doses below 1 Gy for protons and below 0.5 Gy for heavy ions. While low dose-rate experiments are extraordinarily laborious, the challenge is worthwhile, as replicating dose-rates and total dose consistent with the space environment may be worthwhile to help inform future Mars missions.

5.4. Neutrons

Neutron contributions have been considered a primary issue in all workshops and many discussion sessions. Neutrons are secondary particles produced by space radiation interactions with shielding, tissue and other target materials, and there have been only limited radiobiology studies focused on neutron energies of primary importance to space applications, i.e., in the range 1 MeV to 1 GeV. Current radiobiology studies using charged particle beams lack neutron contributions. (The secondary neutrons produced in laboratory exposure rooms make very small contributions to dose.) However, it needs to be clarified that a significant portion of the neutron-induced charged particle field is explicitly represented in the local field approach; namely, the proton elastic recoil and all other nuclear reaction products with Z = 2. The lacking component is the higher LET, short-ranged,

charged tissue target fragments with Z > 2. A neutron spectrum similar to the neutron environment behind shielding in space can be directly produced by stopping a mixed beam of protons and He in an aluminum (or other) target.

Neutron issues include their biological effects, their relevant fluence and energy, and the availability of sources at other facilities. Most of the neutron data are in the few MeV region, whereas space radiation requires studies at the hundreds of MeV up to GeV. Cyclotrons can produce neutrons in the several hundred MeV range and they might provide a useful tool for a limited number of studies to address outstanding biological questions regarding high energy neutrons. An interesting approach to simulating neutrons is to use a mixed, weighted spectrum of hydrogen and helium ions, as such an approach can simulate the effective dose produced by neutrons to within a few percent. On the other hand, experimental biologists are eager to know the direct biological effects of neutrons before replacing with other sources. Neutron measurements are inherently more difficult than charged particles, with the contribution to the dose being less than 10%. (Kohler et al., 2014).

It would be useful to study the differences in biological response between the external field versus local approach. The lack of a complete neutron-induced charged particle field in the local field approach can be a serious issue for biological responses, considering that historically the relative biological effectiveness (RBE) values for neutrons are much higher (up to $50\times$). To disregard that effect on the biology may end up skewing the data obtained from GCR simulation. More studies of the biological effects of neutrons as a function of their energies (especially in the energy region of several hundreds of MeV) are also needed.

5.5. End points

The end points to be studied for cell and animal experiments are crucial. Which end points give the most relevant information for determining oncogenesis, cardiovascular disease, cognitive deficits, etc.? One approach to answering this question is to consider the severity of clustered and persistent DNA damage as a marker for biological response. This severity can be correlated to the measured ionization densities created by the charged particles when traversing biological material. Another approach is to consider cell-to-cell communication in the spread of oxidative damage from targeted cells to neighboring bystander cells.

It may be useful if GCR simulation addresses both cell and animal studies and it may be feasible to design some experiments that allow direct comparison of the *in vitro* and *in vivo* (Kronenberg et al., 2009, 2013) systems. In some cases, the less expensive cell culture experiments may mimic what happens in the body. Validation of cell culture systems with *in vivo* models to confirm or refute the reliability of studies with surrogate systems might increase the number of experimental approaches that can be included in the suite of GCR simulation studies. For studies of cancer incidence, staging, and progression, *in vivo* models remain critical.

5.6. CNS effects

CNS effects pose a particular challenge for space radiobiologists. For example, while quality factor estimates exist for cancer, and RBE estimates are available for degenerative tissue effects, such as cataracts and cardiovascular disease, this information is generally absent for

CNS effects. While quality factors have been considered extensively, and dose equivalent was considered a useful guide for choosing beams for other experiments, such information is not available or useful for CNS studies. Another challenge for CNS studies is that the relevant reference field for CNS may well be different from the Slaba et al. (2015a) field using the local tissue field approach. Time issues are also likely important for CNS effects, as cell repair will occur during non-exposure times. If a cell is hit twice with a year interval between hits, and the damage is sufficient, the cell may die/be removed. However, if the cell survives, some types of cellular damage will likely be repaired, while others will persist depending upon particle type and LET. For example, DNA damage caused by a high-LET Fe ion will be more complex and more difficult to repair than that caused by a low-LET proton. Moreover, epigenetic changes in neurons and glia that alter gene expression and function as a result of ion exposure can persist for long periods of time, and could also be dependent upon particle type. Although there may be one hit per cell every few months for heavy ions, there may be one hit every three days for protons. So, repair molecules will be in use continuously that may result in a sparing effect since they are readily mobilized. Conversely, this scenario, might result in an "inverse" dose-rate effect, i.e. more damage than observed at a high dose-rate, because those repair molecules are already in use repairing damage from a light ion and there is only a limited supply. It is also critical to point out that the geometric cross section of neurons including their many processes is quite large compared to their nuclei and it is currently unknown whether the thin processes and synaptic structures themselves function as the relevant targets. Thus, the "hit per cell" estimates may be systematically underestimated. Finally, there may be spatial coupling between cylindrical particle tracks and three dimensional arrangements of neurons comprising microcircuits leading to simultaneous stimulation/damage of multiple "circuit elements" with enhanced consequences.

5.7. Neurocognitive issues

The move towards a multi-ion beam that covers the spectrum of LETs is a significant step towards modeling the effect of GCR on neurocognitive function. While it will never be possible to deliver a GCR simulation beam in a manner that will reflect the protracted exposure in space, the GCR simulation beam will provide a means to develop a risk assessment model for neurocognitive impairment for a multi-ion exposure. Additional studies will need to be conducted to include the neurocognitive decline that will occur as a result of co-incident neutron exposure. However, given the challenges noted above for ground-based neutron exposure, such effects will have to be treated as additional weighting factors, using co-incident neutron and HZE exposure (achieved via aluminum shielding) to determine the exact weighting factor for neutron exposure. The ability to model the intracranial dose and LET distribution to which astronauts will be subjected – and in particular the likelihood of hot-spots from attenuated particles – should allow for more realistic rodent dose risk assessment studies. Specific GCR simulation beams can be designed that will reproduce similar dose and LET distributions within rats or mice to that observed in humans.

Regardless of GCR simulation or single ion beam exposure, animal age at the time of exposure is an important issue. In its recent NRA (NASA, 2015a), NASA established that rodents for space radiation experiments should be of astronaut-age equivalent at the time of

space radiation exposure (e.g. 5–7 month old mouse is roughly equivalent to a 35 year old human). This issue is expanded in the next section. Also, given the demonstrated dynamic response of the brain and neurocognitive performance to single ion beam exposures, short and long time points after exposure should be evaluated. Finally, given that neurocognitive testing in rodents falls short of mimicking human cognition and behavioral performance, a battery of tests should be employed to gain a full picture of space radiation influence on brain and behavior. Ideally, such tests would have human testing counterparts, such as psychomotor vigilance.

5.8. Extrapolation to humans

There are many open issues, common to both GCR simulation and single beam experiments, in extrapolating animal studies to humans. These include the time scale and the geometry discussed in the next two paragraphs.

The *time scale* of the model systems used needs to be considered. A 3-yr Mars mission lasts about 5% of a human lifetime, but exceeds a mouse lifetime. Scaling the exposure to occur over 5% of a mouse lifetime leads to a dose-rate that is about 20–25 times that experienced by a human on the way to Mars. However, although the biology of aging of the mouse will need to be scaled across its lifetime, the chemistry in the mouse signaling pathways may or may not be the same as in a human, and the reaction rates, at least for initial events, also may be similar to that in humans, leading to an additional modifier to the dose-rate effects. Tissue turnover rates may not scale with lifespan, introducing additional confounding variables. Furthermore, care should be taken in age selection of experimental animals to account for the impact of biochemical or physiological changes at different life stages (e.g. puberty, senescence) on the outcome of radiation effects.

The *geometry* of a mouse irradiation is also affected by the time scale:

- Dose-rate effects that may be caused by interactions between particle signals are not likely to be equally probable in humans and mice.
- The fraction of tissues and organs affected by irradiation at a given fluence will be different in a mouse and a human; the number of cells in 10% of a human brain is equivalent to a much larger percentage of the cells in a mouse brain. To allow as much extrapolation as possible across species, studies should be based on organ doses to be expected in humans, as available from MATROSHKA studies on the ISS (Berger et al., 2013; Reitz et al., 2009). This will help account for the different shielding effect of the tissue surrounding the organs in humans and experimental animals, such as rodents.
- In order to properly interpret the results of experiments intended to simulate space travel, the number of cells responding to the incident particles (i.e., traversed by a track or its delta rays, or receiving chemical signals from other affected cells and tissues) needs to be known or estimated in order to properly scale the experiment.
- At low fluences, the cells responding to an initial incident track are not likely to be the same cells that respond to subsequent tracks, and therefore split low doses

or fractionated low doses are not likely to be an accurate measure of dose-rate effects.

Cancer induction by heavy ions in laboratory animals reveals a relative pattern and can help to understand the mechanisms and the general mode of action. However, absolute extrapolation concerning dose-effect relationships, latency of cancer appearance, time effects and maybe even LET-effect relationships are difficult. Epidemiological studies as proposed for other carcinogens to achieve a more rational risk assessment (Goldsmith et al., 1995) are limited by the very small astronaut population size. Furthermore, the judgment of cancer induction that differs between different animal species with respect to human cancer risk is an unanswered issue. For drugs or chemicals, the distribution in the body and the metabolism (e.g. in the liver) may play a significant role in interspecies differences of cancer induction (Lock and Smith, 2003). For radiation exposure, interspecies differences in the cellular radiation response and in the radiosensitivity of different organs might exist. Studies of both mouse and human cells, especially when performed in environments that mimic the in vivo situation (e.g., oxygen tension, co-culture with other cells typically present in the same microenvironment), may better enable extrapolation from mice to humans. A comparison of rodent and human cellular responses to GCR simulation might support extrapolation of rodent data to humans. This could be achieved through the previously suggested increase in investigator collaboration. Studies with stem cells from tissues as targets for malignant transformation (Trosko et al., 2005) could also help to understand heavy ion induced carcinogenesis.

While rodents have their drawbacks in regards to cancer, they continue to prove an excellent model to study the CNS effects of space radiation, particularly for whole body exposure in the awake and behaving animal. Neurocognitive studies in rodents also provide excellent translational applicability to humans, particularly as many tests in rodents mimic those in humans. Furthermore, given the neuroanatomical, physiological, and biochemical similarities between the rodent and human brain, extrapolations from *in vivo* exposure in rodents has great potential to be extrapolated to humans.

5.9. Beam delivery geometry

GCR incidence in space is isotropic, whereas beam irradiations are axial, thus the geometric distribution of detected radiation events in space may be different from that in a beam. Also, a straight-ahead approximation is not perfectly accurate and one needs to take into account that nuclear reaction products scatter into a relatively large solid angle, thus affecting the distribution of overlapping tracks. Therefore, while GCR simulation studies are worth pursuing, additional 3-dimensional calculations are needed to optimize beam composition.

6. Issues unique to GCR simulation

The issues unique to GCR simulation can be further divided into two groups, namely those issues that arise with the external field approach, versus issues that arise only with the local tissue field approach.

6.1. Multiplicity effects

When a GCR or heavy ion beam strikes an atomic nucleus in a target material, several light particles – including protons or neutrons – can be emitted at the same time (Demoulins et al., 1990; Bastid et al., 1990). These fragments can originate both from projectiles – GCR ions – and targets, which in turn might be the nuclei of the shielding material and/or those of the biological material itself. The number of particles of each kind is called the *multiplicity* of that particle for that reaction.

When the emission of multiple particles (not just protons, but alphas and carbons and others, depending on the mass of the incident projectile) occurs inside a cell (either by a projectile fragmenting on some of the atomic nuclei in the cell, or by multiple pieces coming off an atomic nucleus in the cell that has served as the target of a nuclear reaction), then the simultaneous energy deposition by multiple particles will have a different effect than the same energy deposition occurring over a longer period of time, during which the cell chemistry would have had a chance to return to its equilibrium state. The multiple pieces of a projectile will come out in a narrow angle, and will traverse a large number of neighboring cells at the same time, so that any effects will be correlated. Finally, to the extent that any dose rate effects depend on overlap of tracks, the probability that fragment tracks overlap will be different for fluence randomized over the entire target tissue volume, on the one hand, and for the overlapping of spatially correlated tracks on the other. It is not clear how large such effects almost entirely of projectile fragments (Schimmerling et al., 1989), most of which will be correlated products of heavy ion reactions.

Multiplicity effects are not included explicitly in current modeling calculations, which are based on the use of inclusive cross sections (this accounts for the correct number of particles, but not for any correlations between them). However, the contribution of multiple correlated particles will distort estimates of dose rate (or flux). Multiplicity effects may also influence radiobiological consequences of energy deposition in correlated numbers of cells. The contribution of multiplicity effects would likely seem to be minor, but the magnitude of such effects needs to be estimated with theoretical calculations and Monte-Carlo simulations; if such calculations predict any significant effects, appropriate experiments to test such estimates need to be designed.

6.2. Biological mechanisms

There may exist fundamental differences between the biological mechanisms observed in single beam experiments compared to mixed field experiments of a GCR simulator. As discussed above, synergistic effects may occur, such as a sparing effect due to previously active repair mechanisms, or an inverse dose-rate effect due to resource scarcity from the constant DNA damage and repair cycle, or other physiological changes including transmission of key signaling molecules (or lack thereof), may be especially important for non-cancer effects. A mixed field approach may allow for the validation of existing radiobiological models developed from beam experiments applied to a mixed field environment.

Studies using both single beams and mixed fields would be beneficial, and the body of literature on single beam studies will be useful in interpreting GCR simulation experiments. However, physiological effects in an integrated host environment are more complex and challenging to study, yet are critical to risk estimation. While the issue of biological mechanism applies to both the local tissue field method and the external field method of GCR simulation, there remains the question of difference in biological mechanisms when neglecting high LET tissue target fragments with Z > 2 produced from neutrons or lack of multiplicity in the local field approach compared to the mixed field approach.

If exposures from a GCR simulation are delivered over extended periods of time, another biological phenomenon will need to be considered in the interpretation of the resulting data, namely effects on the cell cycle. It has been established that exposure to both low and high LET radiation can induce cell cycle delay or arrest in some normal cell types (Fournier and Taucher-Scholz, 2004; Pouget and Mather, 2001). For actively dividing cell types with different radio-sensitivity whose cell cycle profiles are altered by radiation exposure to different extents, newly emerging cells over the course of the irradiations will not receive the same number of "hits" over time as others in the population. In contrast, for post-mitotic cells like neurons, multiple doses of ions would be delivered to the same number of cells over time. Because astronauts traveling in space have cell types that would respond in differing manners, both of these phenomena are important. Therefore, comparison of single beams versus mixed fields will be advantageous when interpreting GCR simulation results in terms of cell cycle effects, as well as for cancer mechanisms.

6.3. DNA repair considerations

The complex DNA lesions created after exposure to GCR require various DNA repair processes, and the interplay between the processes remains unknown. This is particularly true for low dose-rate exposures. Likely damage caused by a single GCR particle track includes DNA double- and single-strand breaks, DNA base damage at apyrimidinic/apurinic sites, and DNA-protein crosslinks (Ciccia and Elledge, 2010; Iyama and Wilson, 2013; Asaithamby et al., 2011). Other mutagenic processes will include replication and transcription fork stalling and the emergence of highly persistent RNA:DNA:RNA hybrids (R-loops) (Aguilera and García-Muse, 2012). Repair of these multiple-damaged sites will require simultaneous functioning DNA repair systems, including DNA nucleotide excision repair, base excision repair, single-strand annealing, homologous recombination (HR), nonhomologous end joining (NHEJ), alternative-NHEJ, and newly identified RNA transcription termination factors (RTTFs), including RPRD1B (Kub5/Hera), RPRD1A (p15RS), PSF, XRN2, and SETX (Aguilera and García-Muse, 2012; Helmrich et al., 2013; Morales et al., 2014; Richard et al., 2013). Currently, facility constraints limit our knowledge of: (i) the exact nature of the complex DNA damage created by GCRs; (ii) how such complex lesions are repaired; and most importantly, (iii) what role GCRs play in carcinogenic risk, particularly in gender-specific manners (Morales et al., 2015). Availability of a user-friendly facility for access to GCR exposure is warranted for increased fundamental understanding of DNA damage and repair. In terms of studies using GCR simulation versus single ion beams, the best procedure would be a comparison between the two. This would allow comparison and contrast between the different damages created. For DNA that has undergone repair,

cells can be used that have defects (and then corrections) in any of the repair pathways mentioned above.

6.4. Ordering of beams

Evidence exists (Durand and Olive, 1976; Ngo et al., 1981; McNally et al., 1984; Zhou et al., 2006) that the order in which different beams are delivered may produce different biological outcomes. For example, if one delivers a beam of protons followed by a Si beam, then the biological effect might be different than for a Si beam delivered first, followed by a beam of protons. There are two broad outcomes from multi-beam exposures. Either the beams act independently, which results in an additive effect or the beams interact, which results in an effect that is synergistic. Evidence for both of these outcomes is observed. Synergism is demonstrated with a variety of beam combinations (Durand and Olive, 1976; Ngo et al., 1981; McNally et al., 1984; Yusuke et al., 2004; Zhou et al., 2006). The favored hypotheses is that high-LET radiation damage inhibits or interferes with some aspect of the low-LET repair. Low-LET damage is not repaired as efficiently when combined with higher-LET radiation compared with low-LET radiation alone. Hence, it is not the initial damage which induces differential outcomes but the damage repair response. However, synergy has been shown to depend on the beam order in some cases (Durand and Olive, 1976; Ngo et al., 1981; Zhou et al., 2006), but not in others (McNally et al., 1984). Ngo et al. observe that these differential effects only occur at doses greater than about 8 Gy of neon or X-rays, while Zhou et al. find differences at 40 cGy of iron or titanium combined with protons. Differences in dose-rates (Suzuki, 1993) or interval times as well as radiation or cell type may account for these discrepancies.

Conversely, additivity can be observed with a variety of beam combinations regardless of beam order (Barendsen and Beusker, 1960; Higgins et al., 1983, 1984; Demizu et al., 2004). Interestingly, Higgins et al. demonstrated that it was simultaneous exposure to both photons and neutrons which showed synergistic effects on cell survival, whereas no such effect was observed when beams were delivered sequentially in either order (Higgins et al., 1983). Similar additivity for simultaneous exposures is seen when photons are combined with argon or silicon (Furusawa et al., 2002).

The time interval between exposures is also an important factor to consider when assessing the effects of mixed radiation fields especially in the context of space radiation where doserates for individual ions are very low. There appears to be a window in which beam order dependent effects occur and after which the effects are similar to those calculated from simple additivity of the individual beam effects. (Ngo et al., 1981; McNally et al., 1984; Sutherland et al., 2005; Zhou et al., 2006; Hada et al., 2007). Thus, the risk of a synergistic effect may be substantially lower in a realistic space radiation environment and any radiation scheme which simulates the GCR spectrum should attempt to mimic similar numbers of events which coincide within the window of opportunity for these effects.

If beam order affects biological outcomes, then this phenomenon will not be present in space and one might take steps to avoid it in GCR simulation. A pseudo-random (but repeatable) beam order delivery has been suggested, however, this may not be suitable. In order to obtain a generalization (i.e. probability of some biological outcome or risk), one would have

to repeat the identical experiment many times, with no other changes. This method is not practical or a good use of resources, especially for animals. Comparison of different biological parameters, or with other experiments, end-points, etc. could only be made with many repetitions of each set of conditions. If beam order affects outcomes, this can best be discovered with defined sequences and will be useful information for future aims of low dose-rate experiments. If the order does not matter, there is of course no need for randomization.

7. Applications

One of the aims of the present paper is to list a variety of applications of GCR simulation that may lead to future studies or would benefit from GCR simulation data. Some of these broader applications are now discussed.

7.1. Implementation of a pilot GCR simulation campaign

Initiating a research program devoted to GCR simulation at NSRL is a complicated process. A suggestion is to invite several biologists to submit a realistic research hypothesis. The objective is not to evaluate the merits of each proposed study, but to assume that each has been approved. Each researcher would prepare a submission to NSRL documenting the geometry of the specimen, dose points, and number of specimens or replicates, etc. The NSRL staff would present a menu of ions and energies that represent the GCR reference field, and then assemble a protocol for executing the experiment taking into account the time to physically set up the experiment in the cave, time to switch ions, time to tune energies and duration of exposure for each beam. This would be repeated for each required replicate. The idea is to get a sense of the demands on time and effort and ultimately the cost of establishing a meaningful GCR campaign. An example might be to perform one experimental campaign for carcinogenesis using mice, one study for cardiovascular effects, one study for neurological effects and one study for *in vitro* endpoints of interest. These should be condensed proposals that only emphasize the hypothesis to be tested and irradiation protocols required to test that hypothesis. Emphasis would also be placed on tissue sharing to ensure that the most information is gained from each specimen or animal. This would be an iterative procedure that might require modifications to experiments and compromises for the GCR reference field before hardware and software become finalized at NSRL. Such an approach could also provide guidelines for optimization of experimental design as well as help determine appropriate ions and energies for the GCR reference field.

Rather than immediately implementing the full set of GCR simulation beams and energies, one might start with H, He, O, and Ar. All of these ions share a similar production method at NSRL, which might simplify the ion source configuration. ⁴He and ¹⁶O have essentially the same charge to mass (Q/M) ratio, and it might be possible to produce a mixture of these four ions for simultaneous acceleration and extraction in NSRL. H and He must be included, and the intermediate Z and high Z ions could be selected for other operational reasons.

Another possible consideration that could limit the complexity of the GCR simulator is the question of to what extent experiments can survive a major interruption of a challenging simulation protocol that might take hours or days to fix, such as unanticipated and

unavoidable beam downtime. While this could happen (and has happened) during a single ion campaign at NSRL, the probability of such an occurrence would increase with the complexity of the GCR simulation.

A possible mitigation strategy for the complexity issue would be to initially limit the number of different beams used in the GCR simulator. Concentrating on a limited number of beams and energies would allow the physics and radiobiology to be carefully understood. One advantage of the local field approach in this regard is that it allows for systematic and quantifiable tuning of the simulator complexity while still maintaining a direct connection to the reference field fluence. For the external field approach, it will be important to understand in detail how the radiation field will change after shielding with the most common shielding materials, e.g., Al, polyethylene. For both versions of the GCR simulator, more data are needed for light ion beams, i.e., protons and He. To benchmark and improve the particle and ion transport codes, there is a need to measure differential and double differential cross sections, along with multiplicities. A detailed study using a few beams might also enable more detailed and consistent tests of biological and risk models. Since dose and dose equivalent are not the best risk coefficients, one needs to know the ionization densities created by the charged particles in tissue equivalent materials, and so micro-/nanodosimetric measurements are needed using tissue equivalent proportional counters (TEPC) or compact track structure imaging detectors (Casiraghi et al., 2015).

7.2. GCR simulation using the external field approach

Due to energy limitations at NSRL, much of the present paper has discussed GCR simulation using the local tissue field approach (Slaba et al., 2015a, 2016) instead of the external field approach. However, as previously mentioned, the external field approach may be suitable at higher energy machines such as FAIR (Durante et al., 2007). The present subsection discusses some possible applications of the external field approach.

Using multiple beams designed to simulate the GCR environment in space will provide benchmark data on secondary particle fluences behind shielding materials of varying thicknesses and composition. One example of a possible measurement would be to measure both the fluence and contribution to the total dose and equivalent dose from secondary neutrons. Because of the limitations of neutron instrumentation used in space, space-based measurements of the fluence and dose from neutrons only cover a small range of neutron energies, and as a result the estimates of the total dose from neutrons carry a large degree of uncertainty. Accelerator-based measurements of neutrons can cover a much wider range of neutron energies with a much higher degree of fidelity than can be done in space. With a simulated GCR beam, the production of neutrons and other secondary particles can be sampled simultaneously over a wide range of particle species and energies. When the measurements are conducted under highly controlled and well-defined conditions, as in an accelerator environment, these data can then be used as benchmarks in shielding calculations.

Exposure of materials and electronics to a simulated GCR beam will test the effects of damage due to multiple hits that cannot be performed using a single ion and single ion energy. For example, effects from multiple hits in a material or electronic circuit could be

different depending on whether the multiple hits were due to two protons striking the target area, or a proton and a heavy ion striking the target area. Experiments investigating multiple hits from different ion species would be possible with a GCR simulator. Experiments conducted on shielding materials and electronics are less susceptible to dose-rate effects than biological experiments. Thus, GCR simulator experiments that expose materials to the total dose or fluence expected in a mission should lead to results very close to the same exposures protracted over a longer period of time.

7.3. Biophysical modeling

The GCR simulator will provide data to improve, test, and verify theoretical radiation action and transport models, especially for the track structure and radiation damage modeling community. There are several aspects to consider: theoretical studies should favor a single beam experiment, as it is a pure beam and easier to simulate and to interpret. However, the external field approach introduces radiation contamination to the experimental setup, which is difficult to match in simulations. Local field mixed beams are better characterized for simulation and interpretation, as the contamination is better known. This includes the abovementioned discussion about multiplicities. Furthermore, experiments with a biological endpoint may profit from mixed beam (multiple ions at multiple energies) experiments, as outcomes may be clearer to see and to evaluate as for single beam and energy experiments. This will help refine models and outcome predictions. Another intriguing topic is the possibility to use the GCR simulator as a neutron source. While neutron contribution to the physical dose is mostly understood, their contribution to biological endpoints is still to be examined. Again, experimental data will provide valuable input for the modeling community.

7.4. Simulation of experiments on the International Space Station

The preceding discussion has centered on issues and comparisons between single beam experiments versus GCR simulation experiments conducted at ground-based accelerators. However, space-based radiobiology experiments could benefit significantly from comparisons to ground-based measurements which simulate space conditions. This is possibly an ideal application of GCR simulation. Several radiobiological experiments have been aboard the ISS for some time (Yoshida et al., 2010) and GCR simulation experiments would give valuable information to compare to ISS experiments. However, the best comparisons would need to use the dose and dose-rate found in the ISS, which brings back the issues discussed in previous sections.

8. Current plans for GCR simulator implementation at NSRL

The discussion so far has broadly addressed a range of issues and uses of a GCR simulator, including applications and ideas for implementation. The present section focuses on the actual construction timetable at NSRL, which is of interest to those actually planning to conduct GCR simulation experiments in the near future. Developments are currently underway to provide a GCR simulation beam at NSRL. The GCR Simulator Project at NSRL will generate an accelerator-based spectrum of ions and energies that closely approximate those that are known to make up the shielded GCR environment in space.

These upgrades will enhance the ability to simulate the primary and secondary environment with a mixed field, high-energy capability planned for completion in late 2016. The energy of the final beam line will be upgraded to deliver beams at 1.5 GeV/n, to better represent the energies in the natural GCR. Reference fields will be defined to include the rapid switching of ions such as, protons, He, O, Si and Fe over multiple energies with the possible usage of well-designed absorbers. This will allow GCR species to be simulated with high precision in major energy (or LET) bins. Energy is expected to range from 50-1500 MeV/n with LET ranging between 0.25-1000 keV/µm. To simulate organ exposures within a shielded spacecraft geometry, emphasis will be placed on delivering the majority of the dose from protons (50–60%) and alpha particles (10–20%) with heavier ions (Z > 3) and secondary particles contributing the remainder of the total dose. Validation studies are needed to determine the number of different particles (including particle types and energies) that are sufficient to adequately simulate the biological and health consequences of GCR across major risk areas. An intermediate field capability will be developed to validate final beam selection, including number of species and energies, as well as operational parameters. More detailed descriptions of the field characteristics will be made available on the NSRL website as the upgrades progress.

The plan for implementing GCR simulation at NSRL is to spend 2016 refining the control tools, integrating the various components of the system to deliver a uniform beam of any one of a list of many ions, including protons. The intent is to do so rapidly, reliably and without intervention from machine experts and operators, a feat which will no doubt take more than one iteration. In late 2015 and early 2016, the high energy capability will be tested. The needed upgrades for this were successfully installed during the 2015 summer shutdown. The full GCR capability is expected to be offered in 2017, in both normal and large beam configuration.

The rapid energy changing has been developed at NSRL about eight years ago, and has been undergoing refinements from year to year. It is used routinely for energy changing, taking on the order of 2 minutes to complete in most cases. It is achieved either by scaling the various beam component settings from those used for the current energy to those required for the next, or by restoring archived values, if the desired energy has been already prepared in the past. The advancements which lead to this capability were mostly in the magnet current management software. As for rapid ion changing, it has been under development for the past three years, and required both hardware and software advances. The ion source for the booster is a combination of a Laser Ion Source (LIS) capable of producing a singly charged ion beam from any solid target, a pair of hollow cathode sources, used to do mostly the same with gases, all feeding an Electron Beam Ion Source (EBIS). The EBIS traps the ions and subjects them to bombardment by an intense electron stream which strips them of their own electrons. The time in the trap determines the stripping level. From the EBIS the beam is accelerated through two small LINACs into the booster where it is injected, captured, accelerated and extracted into the NSRL beam line. All the elements mentioned here must be changed in one way or another to affect an ion change. Software tools have been developed that enable this change of settings to take place in an instant. The actual changeover takes between one and two minutes, as certain operations must take place in sequence,

and some active components take time to change setting. Both ion and energy changes are proving to be reliable and repeatable.

This leaves the question of beam quality at the target room. Rapid changes will be of limited use if the beam in the target room requires reshaping every time ions or energies are changed. Some hardware is being added to the extraction magnet, namely a variable-thickness foil-changer, which ensures good control over the spatial distribution of the beam entering the extraction magnet. This, in turn, gives good control over the beam shape in the target room, which is a major factor in the quality of the dose measurement by the ion chamber just upstream from the target stand. When installation is complete, it is expected that both ion and energy changes will be made without affecting the beam shape, and hence calibration, by more than 2–3%, so that these switches should take 1–2 minutes each.

9. Summary and conclusions

A mixed field GCR simulator is an important step toward an understanding of the risks due to space radiation. It is recognized that the GCR simulator will not replace single ion beam investigations, but will serve as a complementary investigation with important implications for the understanding of space radiation risk. The local tissue field approach and the external field approach to the GCR simulator were introduced. Differences and issues, both distinct to each approach to GCR simulation and common to them both, were discussed. A set of conclusions can now be presented.

- In addition to simulating mixed fields, the developments under way to support the GCR simulation beam at BNL are likely to lead to valuable improvements in NSRL usefulness:
 - Fast switching has already been shown to improve throughput of experiments.
 - Generating proton beams independently of the tandem and linear accelerator setup is likely to result in significant operational savings and additional flexibility.
 - Increasing the NSRL ability to handle 1.5 GeV/n Fe will allow more alternatives to obtain mixed fields as well as higher energies for testing and validation of shielding concepts.
 - A second beam line, including modification of the switching magnet so that beams could be switched between beam lines from pulse to pulse would add an important capability, useful for all experiments, with significant cost/benefit improvements.
- GCR simulation studies are worth continuing, but require 3-dimensional calculations to optimize beam composition.
- The local tissue field approach, instead of the external field approach, is most appropriate for NSRL because, due to energy limitations, the external field approach captures only 60% of the effective dose contribution, compared to 90%

for the local tissue field approach for the upgraded NSRL (Slaba et al., 2015a, 2016).

- Hybrid approaches that combine the external field and local tissue field approaches should be pursued to arrive at an optimal beam delivery system that more broadly covers the range of particles, energies, and other factors than either approach independently.
- The effects of neutrons are important and need further study both theoretically and experimentally.
- Experiments at low dose-rate are needed in order to understand the geometry and time scales involved in low fluence exposures.
- Experimental approaches need to be developed to allow identification and counting of cellular targets affected by particle tracks.
- Beam delivery order may be important.
- GCR simulation studies cannot replace single beam experimental programs. Continued single beam experiments are an important approach to understand the biological mechanisms involved in radiation risk and need to be pursued. GCR simulation studies offer an important approach to test and validate mechanistic predictions, as well as provide a possibility of generating new hypotheses.
- Validation studies will be important to determine the number of different particles (including particle types and energies) that are sufficient to adequately simulate the biological and health consequences of GCR across the major risk areas of carcinogenesis, in-flight and late central nervous system effects, as well as cardiovascular and other degenerative tissue diseases from exposure to space radiation.
- Countermeasure testing and validation in a mixed field (high and low LET particles) environment will be a high priority of the GCR simulator.
- The opinions of this paper should be condensed into specified acceptance criteria for a GCR simulator that need to be satisfied by both the prescribed menu of beams and ability of NSRL to implement the protocol.
- A full GCR simulation capability is expected to be available at NSRL in 2017.
- The selected particle types and energies chosen in the mixed field studies should be relatable back to the vast amount of existing radiobiological experimental data collected over the past several decades.

Acknowledgments

Funding for this study was provided by the NASA Human Research Program Space Radiation Element. We are very grateful to Drs. Chris Sandridge, Charles Werneth, Steve Smith, and Jonathan Ransom from NASA Langley Research Center for their careful review of the manuscript.

References

- Aguilar M, et al. AMS Collaboration. Precision measurement of the proton flux in primary cosmic rays from rigidity 1 GV to 1.8 GV with the alpha magnetic spectrometer on the international space station. Phys. Rev. Lett. 2015; 114:171103. [PubMed: 25978222]
- Aguilera A, García-Muse T. R loops: from transcription byproducts to threats to genome stability. Mol. Cell. 2012; 46:115–124. [PubMed: 22541554]
- Asaithamby A, Hu B, Chen DJ. Unrepaired clustered DNA lesions induce chromosome breakage in human cells. Proc. Natl. Acad. Sci. USA. 2011; 108:8293–8298. [PubMed: 21527720]
- Barendsen GW, Beusker TLJ. Effects of different ionizing radiations on human cells in tissue culture. Radiat. Res. 1960; 13:832–840. [PubMed: 13686931]
- Bastid N, Alard JP, Arnold J, Augerat J, Babinet R, Biagi F, Brochard F, Crouau M, Charmensat P, Dupieux P, Fodor Z, Fraysse L, Girard J, Gorodetzky P, Gosset J, Laspalles C, Lemaire MC, Le Merdy A, L'Hôte D, Lucas B, Marroncle J, Montarou G, Parizet MJ, Poitou J, Qassoud D, Racca C, Rahmani A, Schimmerling W, Terrien Y, Valette O. Exclusive measurements of light fragment production at forward angles in Ne–Pb and Ne–NaF collisions at *E/A* = 400 MeV and 800 MeV. Nucl. Phys. A. 1990; 506:637–654. [PubMed: 11537190]
- Berger T, Bilski P, Hajek M, Puchalska M, Reitz G. The MATROSHKA experiment: results and comparison from extravehicular activity (MTR-1) and intravehicular activity (MTR-2A/2B) exposure. Radiat. Res. 2013; 180:622–637. [PubMed: 24252101]
- Cary LH, Ngudiankama BF, Salber RE, Ledney GD, Whitnall MH. Efficacy of radiation countermeasures depends on radiation quality. Radiat. Res. 2012; 177:663–675. [PubMed: 22468705]
- Casiraghi M, Bashkirov VA, Hurley RF, Schulte RW. Characterisation of a track structure imaging detector. Radiat. Prot. Dosim. 2015; 166:223–227.
- Chattopadhyay S. Physics at FAIR. Nucl. Phys. A. 2014; 931:267-276.
- Ciccia A, Elledge SJ. The DNA damage response: making it safe to play with knives. Mol. Cell. 2010; 40:179–204. [PubMed: 20965415]
- Cucinotta F, Chappell L, Kim M. Space radiation cancer risk projections and uncertainties. NASA Technical Paper. 2013:2013–217375.
- Curtis SB. Importance of dose-rate and cell proliferation in the evaluation of biological experimental results. Adv. Space Res. 1994; 14:989–996. [PubMed: 11538040]
- Delgado O, et al. Radiation enhanced lung cancer progression in a transgenic mouse model of lung cancer is predictive of outcomes in human lung and breast cancer. Clin. Cancer Res. 2014; 20:1610–1622. [PubMed: 24486591]
- Demizu Y, Kagawa K, Ejima Y, Nishimura H, Sasaki R, Soejima T, Yanou T, Shimuzu M, Furusawa Y, Hishikawa Y, Sugumura K. Cell biological basis for combination radiotherapy using heavy-ion beams and high-energy X-rays. Radiother. Oncol. 2004; 71:207–211. [PubMed: 15110455]
- Demoulins M, L'Hôte D, Alard JP, Augerat J, Babinet R, Bastid N, Brochard F, Cavata C, DeMarco N, Dupieux P, Fanet H, Fodor Z, Fraysse L, Gorodetzky P, Gosset J, Hayashino T, Lemaire MC, Le Merdy A, Lucas B, Marroncle J, Montarou G, Parizet MJ, Poitou J, Racca C, Schimmerling W, Terrien Y, Valette O. Measurements of a baryon azimuthal emission pattern in Ne + (NaF, Nb, Pb) collisions at 800 MeV per nucleon. Phys. Lett. B. 1990; 241:476–480.
- Durand RE, Olive PL. Irradiation of multi-cell spheroids with fast neutrons versus X rays: a qualitative difference in sub-lethal damage repair capacity or kinetics. Int. J. Rad. Biol. Relat. Stud. Phys. Chem. Med. 1976; 30:589–592.
- Durante M. Space radiation protection: destination Mars. Life Sci. Space Res. 2014; 1:2-9.
- Durante M, Cucinotta FA. Physical basis of radiation protection in space travel. Rev. Mod. Phys. 2011; 83:1245–1281.
- Durante M, Kraft G, O'Neill P, Reitz G, Sabatier L, Schneider U. Preparatory study of a ground-based space radiobiology program in Europe. Adv. Space Res. 2007; 39:1082–1086.
- Fournier C, Taucher-Scholz G. Radiation induced cell cycle arrest: an overview of specific effects following high-LET exposure. Radiother. Oncol. 2004; 73(Suppl. 2):S119–S122. [PubMed: 15971325]

- Furusawa Y, Aoki M, Durante M. Simultaneous exposure of mammalian cells to heavy ions and X rays. Adv. Space Res. 2002; 30:877–884. [PubMed: 12530448]
- Goldsmith DF, Ruble RP, Klein CO. Comparative cancer potency for silica from extrapolations of human and animal findings. Scand. J. Work Environ. Health. 1995; 21(Suppl 2):104–107. [PubMed: 8929704]
- Grahn, D., editor. HZE Particle Effects in Manned Spaceflight. National Academy of Sciences; 1973.
- Hall, EJ., Giaccia, AJ. Radiobiology for the Radiologist. seventh. Lippincott, Williams and Wilkins, Wolter Kluwer; Philadelphia: 2012.
- Hada M, Meador JA, Cucinotta FA, Gonda SR, Wu H. Chromosome aberrations induced by dual exposure of protons and iron ions. Radiat. Environ. Biophys. 2007; 46:125–129. [PubMed: 17237947]
- Hassler, et al. Mars' surface radiation environment measured with the Mars Science Laboratory's Curiosity rover. Science. 2014; 343:1244797. [PubMed: 24324275]
- Helmrich A, Ballarino M, Nudler E, Tora L. Transcription-replication encounters, consequences and genomic instability. Nat. Struct. Mol. Biol. 2013; 20:412–418. [PubMed: 23552296]
- Herr L, Shuryak I, Friedrich T, Durante M, Brenner DJ. New insight into quantitative modelling of DNA double strand break rejoining. Radiat. Res. 2015; 184:280–295. [PubMed: 26305293]
- Heuser JM. The compressed baryonic matter experiment at FAIR. Nucl. Phys. A. 2013; 904–905:941c–944c.
- Higgins PD, DeLuca PM Jr, Pearson DW, Gould MN. V79 survival following simultaneous or sequential irradiation by 15 MeV neutrons and ⁶⁰Co photons. Radiat. Res. 1983; 95:45–56. [PubMed: 6878632]
- Higgins PD, DeLuca PM Jr, Gould MN. Effect of pulsed dose in simultaneous and sequential irradiation of V-79 cells by 14.8 MeV neutrons and ⁶⁰Co photons. Radiat. Res. 1984; 99:591–595. [PubMed: 6473715]
- ICRP, International Commission on Radiological Protection. Assessment of radiation exposure on astronauts in space. ICRP publication 123. Ann. ICRP. 2013; 42(4)
- Isaila M, Schimmerling W, Vosburgh K, White MG, Filz RC, McNulty PJ. Science. 1972; 177:424–425. [PubMed: 17796633]
- Iyama T, Wilson DM. DNA repair mechanisms in dividing and non-dividing cells. DNA Repair (Amst.). 2013; 12:620–636. [PubMed: 23684800]
- Kerr RA. Radiation will make astronauts' trip to Mars even riskier. Science. 2013; 340:1031. [PubMed: 23723213]
- Kim MY, Rusek A, Cucinotta FA. Issues for simulation of galactic cosmic ray exposures for radiobiological research at ground-based accelerators. Front. Oncol. 2015; 5:122. [PubMed: 26090339]
- Kohler J, Zeitlin C, Ehresmann B, Wimmer-Schweingruber RF, Hassler DM, Reitz G, Brinza DE, Weigle G, Appel J, Bottcher S, Bohm E, Burmeister S, Guo J, Martin C, Posner A, Rafkin S, Kortmann O. Measurements of the neutron spectrum on the Martian surface with MSL/RAD. J. Geophys. Res., Planets. 2014; 119:594–603.
- Kronenberg A, Gauny S, Kwoh E, Connolly L, Dan C, Lasarev M, Turker MS. Comparative analysis of cell killing and autosomal mutation in mouse kidney epithelium exposed to 1 GeV/nucleon iron ions in vitro or in situ. Radiat. Res. 2009; 172:550–557. [PubMed: 19883222]
- Kronenberg A, Gauny S, Kwoh E, Grossi G, Dan C, Grygoryev D, Lasarev M, Turker MS. Comparative analysis of cell killing and autosomal mutation in mouse kidney epithelium exposed to 1 GeV protons in vitro or in vivo. Radiat. Res. 2013; 179:511–520. [PubMed: 23560634]
- Little MP, et al. Systematic review and meta-analysis of circulatory disease from exposure to low-level ionizing radiation and estimates of potential population mortality risks. Environ. Health Perspect. 2012; 120:1503–1511. [PubMed: 22728254]
- Lock EA, Smith LL. The role of mode of action studies in extrapolating to human risks in toxicology. Toxicol. Lett. 2003; 140–141:317–322.
- Loeffler JS, Durante M. Charged particle therapy optimization, challenges and future directions. Nat. Rev. Clin. Oncol. 2013; 10:411–424. [PubMed: 23689752]

- McNally NJ, de Ronde J, Hinchliffe M. The effect of sequential irradiation with X-rays and fast neutrons on the survival of V79 Chinese hamster cells. Int. J. Rad. Biol. Relat. Stud. Phys. Chem. Med. 1984; 45:301–310.
- Morales JC, et al. Kub5-Hera, the human Rtt103 homolog, plays dual functional roles in transcription termination and DNA repair. Nucleic Acids Res. 2014; 42:4996–5006. [PubMed: 24589584]
- Morales, JC., et al. RNA transcription termination factors and persistent R-loops: potential carcinogenic determinants after high or low LET IR. The Health Risks of Extraterrestrial Environments (THREE). 2015. https://three.jsc.nasa.gov/#section=main
- NASA. Ground based studies in space radiobiology. NASA Space Radiation Program Element; 2015a. Human exploration research opportunities. Appendix D. NNJ14ZSA001N–radiation.
- NASA. USRA. 2015b. 2015. https://spaceradiation.usra.edu/tissue-sharing/
- NASA. Human research roadmap. 2015c. http://humanresearchroadmap.nasa.gov/
- Ngo FQ, Blakely EA, Tobias CA. Sequential exposures of mammalian cells to low-and high-LET radiations: I. Lethal effects following X-ray and neon-ion irradiation. Radiat. Res. 1981; 87:59–78. [PubMed: 7255673]
- Norbury JW, Slaba TC. Space radiation accelerator experiments the role of neutrons and light ions. Life Sci. Space Res. 2014; 3:90–94.
- Norbury, JW., Slaba, TC., Rusek, A. Galactic cosmic ray simulation at the NASA space radiation laboratory. Space Radiation and Heavy Ions in Therapy Symposium and 7th International Workshop on Space Radiation Research; Osaka, Japan. May 22–24; 2015a.
- Norbury, JW., Slaba, TC., Rusek, A., Durante, M., Reitz, G. International collaboration for galactic cosmic ray simulation at the NASA space radiation laboratory. 15th International Congress of Radiation Research; Kyoto, Japan. May 25–29; 2015b.
- Norbury, JW., Slaba, TC., Rusek, A. Galactic cosmic ray simulator at the NASA space radiation laboratory. 61st Annual Meeting of the Radiation Research Society; Weston, Florida. September 19–22; 2015c.
- Pouget JP, Mather SJ. General aspects of the cellular response to low- and high-LET radiation. Eur. J. Nucl. Med. 2001; 28:541–561. [PubMed: 11357507]
- Richard P, Feng S, Manley JL. A SUMO-dependent interaction between Senataxin and the exosome, disrupted in the neurodegenerative disease AOA2, targets the exosome to sites of transcription-induced DNA damage. Genes Dev. 2013; 27:2227–2232. [PubMed: 24105744]
- Reitz G, et al. Astronaut's organ doses inferred from measurements in a human phantom outside international space station. Radiat. Res. 2009; 171:225–235. [PubMed: 19267549]
- Schimmerling W, Miller J, Wong M, Rapkin M, Howard J, Helmut JH, Spieler G, Jarret BV. The fragmentation of 670A MeV neon-20 as a function of depth in water. I. Experiment. Radiat. Res. 1989; 120:36–71. [PubMed: 2798782]
- Silver L. Physics and biophysics experiments needed for improved risk assessment in space. Acta Astronaut. 2008; 63:886–898.
- Slaba TC, Blattnig SR. GCR environmental models I: sensitivity analysis for GCR environments. Space Weather. 2014; 12:217–224.
- Slaba TC, Blattnig SR, Norbury JW, Rusek A, La Tessa C, Walker SA. GCR simulator reference field and a spectral approach for laboratory simulation. NASA Technical Paper. 2015a:2015–218698.
- Slaba, TC., Blattnig, S., Norbury, JW., Rusek, A., La Tessa, C., Walker, S. GCR simulator reference field and a preliminary beam selection strategy at the NASA space radiation laboratory. 61st Annual Meeting of the Radiation Research Society; Weston, Florida. September 19–22; 2015b.
- Slaba TC, Blattnig SR, Walker SA, Norbury JW. GCR simulator reference field and a spectral approach for laboratory simulation. Life Sci. Space Res. (accompanying paper in this issue). 2016
- Stohlker T, Bagnoud V, Blaum K, Blazevic A, Braunin-Demain A, Durante M, Herfurth F, Lestinsky M, Litvinov Y, Neff S, Pleskac R, Schuch R, Schippers S, Severin D, Tauschwitz A, Trautmann C, Varentsov D, Widmann E. APPA at FAIR: from fundamental to applied research. Nucl. Inst. Methods Phys. Res. B. 2015; 365:680–685.
- Sutherland BM, Cuomo NC, Bennett PV. Induction of anchorage-independent growth in primary human cells exposed to protons or HZE ions separately or in dual exposures. Radiat. Res. 2005; 164:493–496. [PubMed: 16187755]

- Suzuki S. Survival of Chinese hamster V79 cells after irradiation with a mixture of neutrons and 60 Co γ rays: experimental and theoretical analysis of mixed irradiation. Radiat. Res. 1993; 133:327–333. [PubMed: 8451382]
- Trosko JE, Chang CC, Upham BL, Tai MH. Low-dose ionizing radiation: induction of differential intracellular signalling possibly affecting intercellular communication. Radiat. Environ. Biophys. 2005; 44:3–9. [PubMed: 15821925]
- Ullrich RL, Jernigan MC, Storer JB. Neutron carcinogenesis: dose and dose-rate effects in BALB/c mice. Radiat. Res. 1977; 72:487–498. [PubMed: 339261]
- Walker SA, Townsend LW, Norbury JW. Heavy ion contributions to organ dose equivalent for the 1977 galactic cosmic ray spectrum. Adv. Space Res. 2013; 51:1792–1799.
- White MG, Isaila M, Prelec K, Allen HL. Heavy ion acceleration. Science. 1971; 174:1121–1123. [PubMed: 17779398]
- Yoshida K, Yoshida S, Kasai K, Morita T. Study of the effects of space radiation on mouse ES cells. Biol. Sci. Space. 2010; 14:11–15.
- Yusuke D, et al. Cell biological basis for combination radiotherapy using heavy ion beams and high energy X-rays. Radiother. Oncol. 2004; 71:207–211. [PubMed: 15110455]
- Zeitlin C, et al. Measurements of energetic particle radiation in transit to Mars on the Mars Science Laboratory. Science. 2013; 340:1080–1084. [PubMed: 23723233]
- Zhou G, Bennett PV, Cutter NC, Sutherland BM. Proton-HZE-particle sequential dual-beam exposures increase anchorage-independent growth frequencies in primary human fibroblasts. Radiat. Res. 2006; 166:488–494. [PubMed: 16953667]